Author’s response to reviews

Title: Esophageal adenocarcinoma with leukemoid reaction: A case report

Authors:
Ge Yu (whxwkyg@163.com)
Huai-jun Ji (sdhzcxjhj@sina.com)
Chuizheng Meng (whxwkmcz@126.com)
Yixuan Huang (whxwkhyx@sina.com)
Guogang Gao (whxwikgg@sina.com)
Chuanping Liu (whxwiklc@163.com)
Shanlei Wang (whxwkwsl@sina.com)
Lei Zhang (whxwklzl@126.com)
Jin Ju (whxwikjj@sina.com)

Version: 1 Date: 20 Mar 2019

Author’s response to reviews:

2019.03.20 20:50

Editorial Office,

I'm sorry for the trouble I caused. I have made modification as the required. Thank you.

With kind regards,

Best wishes,

Huaijun Ji

2019.03.19 21:49

Editorial Office,
In a later revision of the article, we amended Jin Ju as co-corresponding author, but there is no way to label co-corresponding authors in the submission system. I have checked the list of authors in my manuscript file and the submission system, the name and order are the same.

With kind regards,

Best wishes,

Huaijun Ji

Dear Dr Vipin Zamvar,

Thank you for your email. I am pleased that my article is potentially acceptable for publication in Journal of Cardiothoracic Surgery. I have revised the article as your request point-by-point.

1. There are some English mistakes should be revised.

We've had polished the language by professional institutions (Editage). Thank you.

2. The leukocytosis is a well known condition that can be encountered with different tumors in the body, moreover, it has been reported with squamous cell esophageal carcinoma and was linked to prognosis and outcome. I think authors should mention this clearly in their manuscript.

I've added the content about squamous cell esophageal carcinoma and leukocytosis. Thank you.

3. Moreover, there is no systematic review of literature concerning the subject hence the title should be a case report and eliminate the review of literature to be more precise.

The title of the article was revised to “Esophageal adenocarcinoma with leukemoid reaction: A case report”. Thank you.

4. Finally, concerning the references, authors should revise them as the authors in the section of discussion referred to Ganger et al by number (16) and reference number (16) at the order of references referred to HurtadoCordovi J.

I have corrected this error in the article. Thank you.
5. Jin Ju contributed greatly to the revision of the article. With the consent of all authors, we amended him as co-corresponding author. We hope that this request meets with your approval. Thank you.

Best wishes,

Huaijun Ji

sdhzcxjhj@sina.com